Author's response to reviews

Title: The Role of a Clinical Nurse Consultant in an Australian Health District

Authors:

Lesley Wilkes (l.wilkes@uws.edu.au)
Lauretta Luck (l.luck@uws.edu.au)
Jennifer O'baugh (j.obough@health.nsw.gov.au)

Version: 2 Date: 14 December 2014

Author's response to reviews: see over
Sunday, 14 December 2014

Executive Editor
BMC Nursing
BioMed Central
236 Gray’s Inn Road
London WC1X 8HB
United Kingdom

Dear Editor-in-Chief,

RE: Revision of manuscript for the paper titled: The Role of a Clinical Nurse Consultant in an Australian Health District.

Thank you for the swift feedback on missing items on our manuscript. We have amended sections and detailed these changes in the points below.

- Addition of Title page (line 1-35)
- Change of the title to include study design (line 3)
- Addition of acknowledgements (line 59-62)

We look forward to hearing from you.

Sincerely,

Professor Lesley Wilkes